Author's response to reviews

Title: Yellow fever control in Cameroon: where are we now and where are we going?

Authors:

Charles Shey Wiysonge (charles.wiysonge@mrc.ac.za)
Emmanuel Nomo (a_nsina@yahoo.fr)
Jeanne N Mawo (ngo_ndjan@yahoo.com)
James Ofal (ofalutia@yahoo.com)
Julienne Mimbouga (wiysonge@yahoo.com)
Johnson Ticha (tichajohnson@yahoo.fr)
Peter M Ndumbe (pndumbe@yahoo.com)

Version: 3 Date: 24 January 2008

Author's response to reviews:

Charles Shey Umar Wiysonge, MD
Senior Scientist,
South African Medical Research Council
Francie van Zyl Drive, Cape Town
South Africa
Tel: +27 (21) 938 0508
Fax: +27 (21) 938 0836
Email: charles.wiysonge@mrc.ac.za

24 January 2008

Dr Annabel Phillips
Senior Assistant Editor
BMC-series Journals
BioMed Central

Re: MS: 1825222038162597
Yellow fever control in Cameroon: where are we now and where are we going?

Dear Annabel,

Thank you for accepting, in principle, to publish our manuscript in BMC Medicine. We are very excited about this.

We have now added a new paragraph in the Discussion section (paragraph 2) on adverse events following yellow fever immunisation, as suggested by the peer-reviewer. This addition has improved our manuscript further, and we are
grateful to the reviewer for the suggestion.
We have also formatted the revised manuscript to conform to BMC Medicine’s style.
We trust that the revision meets the requirements for final acceptance of the manuscript in BMC Medicine.
Thank you for your time and consideration

Yours sincerely,

Charles